Author’s response to reviews

Title: Timing of surgery for hip fracture and in-hospital mortality. A retrospective population-based cohort study in the Spanish National Health System.

Authors:

Julián Librero (julian.librero@uv.es)
Salvador Peiró (peiro_bor@gva.es)
Edith Leutscher (ed.leutscherv@coma.es)
Juan Merlo (Juan.Merlo@med.lu.se)
Enrique Bernal-Delgado (ebernal.iacs@aragon.es)
Manuel Ridao (ridao_man@gva.es)
Natalia Martínez-Lizaga (nmartinez.iacs@aragon.es)
Gabriel Sanfélix-Gimeno (sanfelix_gab@gva.es)

Version: 5 Date: 14 January 2012

Author’s response to reviews:

Author list: Julián Librero, Salvador Peiró, Edith Leutscher, Juan Merlo, Enrique Bernal-Delgado, Manuel Ridao, Natalia Martínez and Gabriel Sanfélix-Gimeno

Title : Timing of surgery for hip fracture and in-hospital mortality. A retrospective population-based cohort study in the Spanish National Health System.

Journal: BMC Health Services Research

MS ID : 1546736131528395

Major revisions (we require the author to make these changes)

Author groups: please include the name of the group within the list of authors, by adding the name to the `first name¿ box; and the bracketed abbreviation to the `last name¿ box in the submission system. Please remove the names of the members of the group from the title page and place them in the Acknowledgements, as they are to be acknowledged as members of the study group. The group’s contribution as a whole can be given in the authors contributions section.

- The VPM-SNS Group members that are not included nominally in the list of authorship should not be considered authors of the paper. The role of these members in this article is minor (limited to general aspects of data collection from the original data bases), do not have reviewed or approved the final version and we think this work is insufficient to consider the authorship of all members the group.

- Papers from the VPM-SNS group usually include an annexe with the list of their members because there is always some degree of collaboration, but without involving considerations of authorship.

- To avoid confusions, we have finally opted for: 1) to remove the VPM-SNS
Group from the author’s list, and the list of their members in the acknowledgments section, 2) to include the VPM-SNS group between the affiliations of the corresponding authors.

Competing Interests: Please state that the authors do or do not have any competing interests in this section.
- We have included the corresponding statement.
- Additionally, we have moved the discharge paragraph to the acknowledgments section.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Tracking: Please remove all highlighting and tracking from the manuscript. and please ensure that all text in the manuscript is black.
- We have removed all highlighting and tracking from the manuscript, and we have changed to black the text in any other color.